Author's response to reviews

Title: Association of obesity risk SNPs in PCSK1 with insulin sensitivity and proinsulin conversion

Authors:

Martin Heni (martin.heni@med.uni-tuebingen.de)
Axel Haupt (HAUPT_AXEL@lilly.com)
Silke A. Schäfer (silke.schaefer@med.uni-tuebingen.de)
Caroline Ketterer (caroline.ketterer@med.uni-tuebingen.de)
Claus Thamer (claus.thamer@med.uni-tuebingen.de)
Fausto Machicao (fausto.machicao@med.uni-tuebingen.de)
Norbert Stefan (norbert.stefan@med.uni-tuebingen.de)
Harald Staiger (harald.staiger@med.uni-tuebingen.de)
Hans-Ulrich Häring (hans-ulrich.haering@med.uni-tuebingen.de)
Andreas Fritsche (andreas.fritsche@med.uni-tuebingen.de)

Version: 3 Date: 8 June 2010

Author's response to reviews:

Dear Sir or Madame,

Thank you for accepting our manuscript for publication in BMC med genetics.
We did all formatting changes requested by the BMC Series editorial production team.

Best regards,
Martin Heni

#############################

Major revisions (we require the author to make these changes)

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
We now state this in the manuscript.

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend, preferably indicated with bold text.
Figure titles are given in bold now.

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system. Please ensure all files are referred to as additional files (not, for example, supplementary data) in all citations and the additional file descriptions.
We removed all additional files.

Thank you!

Best regards,

Martin Heni